Author’s response to reviews

Title: Effectiveness of Trauma Team on Medical Resource Utilization and Quality of Care for Patients with Major Trauma

Authors:

Chih-Jung Wang (poemcage@gmail.com)
Shu-Ting Yen (j_yen33@ymail.com)
Shih-Fang Huang (cycc0621@gmail.com)
Su-Chen Hsu (suchenhsu@gmail.com)
Jeremy Ying (jying@isu.edu.tw)
Yan-Shen Shan (ysshshan@mail.ncku.edu.tw)

Version: 3 Date: 02 Jul 2017

Author’s response to reviews:

20170702

Dear Editor:

The following is the response to critique of submitted manuscript: BHSR-D-16-01102R2; Effectiveness of Trauma Team on Medical Resource Utilization and Quality of Care for Patients with Major Trauma. Chih-Jung Wang; Shu-Ting Yen; Shih-Fang Huang, BS; Su-Chen Hsu; Jeremy C Ying, PhD; Yan-Shen Shan

1) In the Consent to publish section, this section is for if patients have given consent to publish their individual details. As this is not applicable to your study, please state "Not applicable". 1) Please note, the "Availability of Data and Materials" declaration refers to the raw data used in your study. Presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section.
If you do not wish to share your data, please clearly state this in this section along with a justification. For more information and a list of suitable availability statements, please see our Submission guidelines:

https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

-->I have stated that “The data is available at registry bank of Division of Trauma, Department of Surgery, National Cheng Kung University Hospital. It will be available upon request to the corresponding/first author. (page 9-page 10).

2) In your "Ethics and Consent to Participate" declaration, along with the information currently provided, please state whether informed consent was provided by all patients/the data owner. If the need for consent was waived by the ethics committee, or is not required due to national legislation, please state this, with a reference if appropriate.

-->I have added “The demographic and clinical information for the patients were de-linked prior to analysis. Thus, informed consent was not obtained from patients prior to the study.” In both the section of Material and Method (page 5-page 6) and page 9.

Yan-Shen Shan MD,PhD
Distinguished Prof and Director
Department of Clinical Medicine Research and Surgery, NCKUH
Institute of Clinical Medicine, College of Medicine, NCKU
ysshan@mail.ncku.edu.tw